Author's response to reviews

Title: Paracardial pancreatic pseudocyst with isolated thoracic symptoms: a case report

Authors:

Robert Drescher (robert.drescher@rub.de)
Odo Koester (odo.koester@rub.de)
Carsten Lukas (carsten.lukas@rub.de)

Version: 2 Date: 27 August 2007

Author's response to reviews:

Dear Ms Shaw,

attached you will find the revised version of our manuscript. The title was changed, the blind title page removed, and there are now three single figures.

Thank you very much!

Kind regards,

Robert Drescher

Dear Dr Drescher,

Journal of Medical Case Reports MS: 1782734468155553
Paracardial pancreatic pseudocyst causing isolated thoracic symptoms
Robert Drescher, Odo Koester and Carsten Lukas

Thank you for submitting the above manuscript to Journal of Medical Case Reports.

There are a number of matters that require your attention before your manuscript can be sent to the Editorial Board for initial consideration, as the manuscript does not conform to the journal style. These are outlined below, but I advise that you check our instructions for authors(http://jmedicalcasereports.com/author/manuscript/case/information/view.do) for further information:

Restructure the title so it ends in ':a case report.'
Remove the 'Bling title' page
Remove one figure as it is journal policy to allow a maximum of 3 figures.
Please could you modify the manuscript and submit a revised version by logging on to the website http://www.biomedcentral.com/my/manuscripts. We would
encourage you to do so as soon as possible in order to minimize the delay to the review of your manuscript.

Your manuscript will not be sent for peer review until the revised manuscript has been received.

If you have any queries please don’t hesitate to get in touch.

Best wishes,
Claire Shaw
Assistant Editor - Independent Journals